Author's response to reviews

Title: Patient-physician interaction in general practice and health inequalities in a multidisciplinary study: design, methods and feasibility in the French INTERMEDE study.

Authors:

Michelle Kelly-Irving (kelly@cict.fr)
Christine Rolland (rolland@cict.fr)
Anissa Afrite (Afrite@irdes.fr)
Chantal Cases (Cases@irdes.fr)
Pierre Dourgnon (Dourgnon@irdes.fr)
Pierre Lombrail (pierre.lombrail@chu-nantes.fr)
Jean Pascal (jean.pascal@chu-nantes.fr)
Thierry Lang (lang@cict.fr)

Version: 2 Date: 10 April 2009

Author's response to reviews:

I think I have made all the amendments requested, except for one regarding reference 17. I am not sure how to reference this document differently, as it is the ethical approval document and not an article. It may not be necessary to reference it at all, and I will leave it to your discretion as to whether to leave it in or remove it from the article. Many thanks.